Author's response to reviews

Title: Worksite interventions for preventing physical deterioration among employees in job-groups with high physical work demands: Background, design and conceptual model of FINALE

Authors:

Andreas Holtermann (aho@nrcwe.dk)
Marie B Jørgensen (mbj@nrcwe.dk)
Bibi Gram (bgram@health.sdu.dk)
Jeanette R Christensen (jrc@sport.au.dk)
Anne Faber (afh@nrcwe.dk)
Kristian Overgaard (ko@sport.au.dk)
John Ektor-Andersen (ektor-andersen.john@telia.com)
Ole S. Mortensen (osm@nrcwe.dk)
Gisela Sjøgaard (gsjogaard@health.sdu.dk)
Karen Søgaard (ksogaard@health.sdu.dk)

Version: 7 Date: 5 March 2010

Author's response to reviews:

The abstract present in the manuscript does now match exactly what is in the submission system